Author’s response to reviews

Title: Online, paper or mixed? A randomised trial and economic evaluation of the effect of response mode on response rate, response bias, and item non-response in a survey of doctors

Authors:

Anthony Scott (a.scott@unimelb.edu.au)  
Sung-Hee Jeon (shjeon@unimelb.edu.au)  
Catherine M Joyce (catherine.joyce@med.monash.edu.au)  
John Humphreys (ohn.Humphreys@monash.edu)  
Guyonne Kalb (g.kalb@unimelb.edu.au)  
Julia Witt (witt@cc.umanitoba.ca)  
Anne Leahy (anneel@unimelb.edu.au)

Version: 2  Date: 24 November 2010

Author’s response to reviews:

Dear Editors,

As suggested in your email of 10 Nov 2010, we would like to accept your offer to arrange for the transfer of this manuscript to BMC Medical Research Methodology.

Your email indicated that we should provide information about trial registration. Please note that although we have conducted a randomised controlled trial, it is not a clinical trial nor a trial of a health intervention of any kind, and therefore registration is not required.

We have provided 2 additional files for your information:

1. CONSORT flow diagram
2. CONSORT checklist

The manuscript has been changed to include a reference to the flow chart as Figure 1.

regards

Catherine Joyce